Author’s response to reviews

Title: Effects of high intensity interval training on exercise capacity in people with chronic pulmonary conditions: a narrative review

Authors:

Abbey Sawyer (Abbey.Sawyer@health.wa.gov.au)

Vinicius Cavalheri (vinicius.cavalheri@curtin.edu.au)

Kylie Hill (K.Hill@curtin.edu.au)

Version: 2 Date: 22 Feb 2020

Author’s response to reviews:

The corresponding author of the manuscript is Dr Kylie Hill. This has been adjusted.